Author's response to reviews

Title: Assessment of the impact of the scanner-related factors on brain morphometry analysis with Brainvisa

Authors:

Mahsa Shokouhi (m.shokouhi.1@research.gla.ac.uk)
Anna Barnes (ab698@cam.ac.uk)
John Suckling (js369@cam.ac.uk)
Thomas W.J. Moorhead (bill.moorhead@ed.ac.uk)
David Brennan (d.brennan@clinmed.gla.ac.uk)
Dominic Job (Dominic.Job@ed.ac.uk)
Katherine Lymer (katherine.lymer@ed.ac.uk)
Paola Dazzan (paola.dazzan@iop.kcl.ac.uk)
Tiago Reis Marques (tiago.marques@kcl.ac.uk)
Clare MacKay (clare.mackay@psych.ox.ac.uk)
Shane McKie (shane.mckie@manchester.ac.uk)
Steven C. R. Williams (Steve.Williams@iop.kcl.ac.uk)
Stephen M. Lawrie (s.lawrie@ed.ac.uk)
Bill Deakin (bill.deakin@manchester.ac.uk)
Steve R. Williams (steve.williams@manchester.ac.uk)
Barrie Condon (Barrie.Condon@glasgow.ac.uk)

Version: 4 Date: 20 December 2011

Author's response to reviews: see over
Dear editorial board of BioMed Central,

We are thankful to the BMC Series Editorial Production Team for their comments and requesting for formatting changes to the manuscript. We have modified the manuscript accordingly. The explanation of what we have changed in accordance with the requirements is given point by point in the following page.

We thank you in advance for your kind consideration.

Sincerely yours,

Mahsa Shokouhi on behalf of the authors
Major revisions

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

The authors' list appears exactly the same in the manuscript and the submission system and the style is modified according to the above.

Equations: Please note that the PDF version of articles in this journal is laid out in a two-column format. We therefore request that you break long equations so that they can be rendered legibly within a 85mm column width.

The font of the equation has been changed to reduce its width to the required limit.

Minor revisions

Tables: please remove the visible vertical lines from your tables

The vertical lines are now removed.

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements

The merged cells (merging cells of two different rows) have been removed.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

The manuscript has been checked for typographical errors.